Notes of the RICH1 Phone meeting 19/01/2007
People Present:
CERN:  Clara(part), Fabio, Olav, Richard P.
IC:         Bill, Dave, Gabriel(part), Trevor(part)
Oxford:  Rohan(part), Neville, Matthew, Joe
Apologies for absence:
Mike, Nick
Discussed Agenda Topics:
1. Gas Enclosure
Before Xmas the Seal to VELO at the GE end was released to allow for the VELO rotation. The operation was successful and the VELO was rotated. The first installation of the GE side doors with the crane was carried out successfully. Now the GE is closed up. When the beampipe is aligned the seal to the exit window can be completed (action Derek/Peter), the exit window bolts tightened and the leak tests can be done (~Feb07).

2. Quartz Windows
Three QW have been coated. One needs to be done. The 4 QW will be ready by the end of January and Kenneth will measure the transmission (action Kenneth).

3. Spherical Mirrors
CMA January 12th acceptance: Last week Geoff visited CMA and participated in the assembly of the 4 mirrors. Three mirrors had the surface finish complete. The D0 appears good (D0~1mm). One mirror needed re-surfacing (small area ~0.1%) because of the bad surface near the beampipe cut-out. The vertical gap between the mirrors was fine while the horizontal gap needed correcting.  The mirror alignment testing worked fine. The mounts and outside corners were shaved off to reduce the mirror diagonal for the coatings at SESO.  The frame plus the 4 mirrors will be shipped end of next week (~1-2 weeks delivery). CMA phone meeting is planned for next Wednesday.
Coating at SESO: SESO (France) sent an updated quote (same price) for step2 which needs to be put through EDH. The dates proposed by SESO (7 Feb) need to be discussed because it is unlikely to have the mirrors ready by then (action Fabio). The last analyses on the contamination samples by SESO indicate that there was a process anomaly which caused a rise up of the pumping oil in the chamber during contamination tests with the CMA samples, resulting in false outgassing of the CMA samples. SESO will put contamination samples during the coatings of the CMA mirrors to be on the safe side.  
Mirror Installation tooling: Drawings completed. Cutting and welding has started. Delivery ~ end Feb07
4. Flat Mirrors
Optical alignment setup: Matthew has completed the model design and the material has been ordered. It should be ready in 4-6 weeks. The drawings will be put on EDMS (action Fabio).
.
Installation Boxes: Richard’s drawings have been given to the CERN “Sous Traitance” service. The quote will be ready next week. The 4 boxes should be ready by the end of Feb07.
5. HPD Assembly
An HPD review meeting is planned at IC for the second week in February. 
Dressing: Six of the eight ladders have been assembled at CERN. The end pieces need to be anodized. First a spare RICH2 column will be built and then the dressing of the RICH1 ladders will go ahead.  
Shields: There are enough shields for the first half of RICH1. The shields have been tested and are fine (very low failure rate). The remaining ones are being tested.
Cooling Plates: Received only 14 pipes out of 32 (company mistake). Only 14 pipes are still at the company. The missing pipes will be taken from the RICH2 leftovers. The copper plates are slightly out of dimensions (not a problem).

Rail System: Worked on how to handle the installation box with the minimum amount of handling material. The materials will be ordered next week and will be manufactured in house, apart from welding that needs to be certified for CERN safety requirements.

Cabling: Cabling is coordinated by Ken and Rolf. The cable trays from the balcony to the lower part of RICH1 have been installed, while the cable trays to the top of RICH1 stop at the tunnel and will be extended when the scaffolding is mounted for bringing in the cables. The cables are connected in situ.   

6.  MDMS
Ray gave a presentation Wed17 Jan07. It is not clear if the proposed system will fit. Expect a definite answer in ~ two weeks at the next MDMS phone meeting.
7. LAMS

The cameras will be bought by Antonis and Paul S. should have the fibers (action Gabriel to coordinate).

.
8. Aerogel
To be installed as last ~August 07. A test assembly is planned for March.

9. Next meeting
Friday 2nd February 2007 at 10:00(GMT) 11:00(CET). 
